Author's response to reviews

Title: Work-related psychosocial events as triggers of sick leave - results from a Swedish case-crossover study.

Authors:

Hanna Hultin (hanna.hultin@ki.se)
Johan Hallqvist (johan.hallqvist@ki.se)
Kristina Alexanderson (kristina.alexanderson@ki.se)
Gun Johansson (Gun.Johansson@liu.se)
Christina Lindholm (Christina.Lindholm@ki.se)
Ingvar Lundberg (Ingvar.Lundberg@medsci.uu.se)
Jette Möller (jette.moller@ki.se)

Version: 4 Date: 26 January 2011

Author's response to reviews:

To the editorial production team:

We have made the requested changes in the manuscript. Furthermore we have made some typographical corrections. We hope you will find this satisfactory. Please, do not hesitate to contact us if you have any questions.

Kind regards,
Hanna Hultin
on behalf of the authors